

Author's response to reviews

Title: Prospective study of urinary tract infection surveillance after kidney transplantation

Authors:

Roberto Ribera-Sanchez (cerdea321@yahoo.com.mx)
Dolores Delgado-Ochoa (lolgera@hotmail.com)
Ramon Espinoza-Hernández (espinohdzt2003@yahoo.com.mx)
Rocio Flores-Paz (rflorespaz@yahoo.com.mx)
Elvia García-Jiménez (elviagen@yahoo.com.mx)
Andres Bazan-Burges (bazandr@hotmail.com)
Myriam Arriaga-Alba (arriaga_alba@yahoo.com)

Version: 10 **Date:** 2 July 2010

Author's response to reviews:

Dear Sirs;

This new version is in fact the same we have sent you on June 18th. The only changes done to the document are English editor. We have sent the document to <http://www.internationalscienceediting.com/>,

The Ethical considerations section has been re-edited. Some minor editorial changes have been done on conclusion under the abstract and those conclusions at the end of the document.

We are including two documents.

The main manuscript is 2010 BMC CHANGES ACCEPTED.doc. We are also including a document 2010 BMC Transplant_revised.doc, where the changes are remarked in color.

Yours faithfully;

Myriam Arriaga Alba.